Author's response to reviews

Title: Prevalence of asthma and allergies in the Greek-Cypriot and Turkish-Cypriot communities in Cyprus: a bi-communal cross-sectional study.

Authors:

Demetris Lamnisos (demetris.lamnisos@cut.ac.cy)
Maria Moustaki (mar.moustaki@gmail.com)
Ourania Kolokotroni (ourania.kolokotroni@cut.ac.cy)
Huseyin Koksoy (hkoksoy@yahoo.com)
Muharrem Faiz (kadem@mycyprus.net)
Kenan Arifoglu (karifoglu@yahoo.com)
Donald K Milton (dmilton@umd.edu)
Nicos Middleton (nicos.middleton@cut.ac.cy)
Panayiotis K Yiallouros (p.yiallouros@cut.ac.cy)

Version: 2 Date: 28 September 2012

Author's response to reviews: see over
28 September 2012

BMC Public Health
Editorial Office

Dear Editor,

Re: Prevalence of asthma and allergies in the Greek-Cypriot and Turkish-Cypriot communities in Cyprus: a bi-communal cross-sectional study.

I am writing to submit a revision of the above manuscript (MS:1199178007804063) that was requested by the Journal Editorial Office (Mr Victorino Silvestre).

Thank you for noticing in the manuscript the lack of a statement on the Ethics Committee approval of this study. In this regard in the “Methods” Section of the manuscript and in Sub-section “Study Population and design” we have now added a reference on the ethics committee approval of the study that reads as follows: “The study was approved by the Cyprus Bioethics Committee (EEBK/EP/2006/28).”

Thank you.

Yours sincerely,

Panayiotis Yiallouros M.D., PhD